Use of Analog and Human Insulin in a European Hemodialysis Cohort With Type 2 Diabetes: Associations With Mortality, Hospitalization, MACE, and Hypoglycemia

Thomas Ebert, MD, Nosheen SattarMD, Marni Greig, PhD, Claudia Lamina, PhD, Marc Froissart, PhD, Kai-Uwe Eckardt, MD, Jürgen Floege, MD, Florian Kronenberg, MD, Peter Stenvinkel, MD, David C. Wheeler, MD, James Fotheringham, PhD

PII: S0272-6386(23)00775-8

DOI: https://doi.org/10.1053/j.ajkd.2023.05.010

Reference: YAJKD 57952

To appear in: American Journal of Kidney Diseases

Received Date: 7 November 2022

Revised Date: 17 May 2023

Accepted Date: 23 May 2023

Please cite this article as: Ebert T, SattarMD N, Greig M, Lamina C, Froissart M, Eckardt KU, Floege J, Kronenberg F, Stenvinkel P, Wheeler DC, Fotheringham J, Use of Analog and Human Insulin in a European Hemodialysis Cohort With Type 2 Diabetes: Associations With Mortality, Hospitalization, MACE, and Hypoglycemia, *American Journal of Kidney Diseases* (2023), doi: https://doi.org/10.1053/j.ajkd.2023.05.010.

This is a PDF file of an article that has undergone enhancements after acceptance, such as the addition of a cover page and metadata, and formatting for readability, but it is not yet the definitive version of record. This version will undergo additional copyediting, typesetting and review before it is published in its final form, but we are providing this version to give early visibility of the article. Please note that, during the production process, errors may be discovered which could affect the content, and all legal disclaimers that apply to the journal pertain.

© 2023 Published by Elsevier Inc. on behalf of the National Kidney Foundation, Inc.



Use of Analog and Human Insulin in a European Hemodialysis Cohort With Type 2 Diabetes: Associations With Mortality, Hospitalization, MACE, and Hypoglycemia

Thomas Ebert, MD^{1,2},* Nosheen SattarMD³,* Marni Greig PhD^{3,4}, Claudia Lamina, PhD⁵, Marc Froissart, PhD⁶, Kai-Uwe Eckardt, MD⁷, Jürgen Floege, MD⁸, Florian Kronenberg, MD⁵, Peter Stenvinkel, MD¹, David C Wheeler, MD⁹, and James Fotheringham, PhD^{10,11}

1: Division of Renal Medicine, Department of Clinical Science, Intervention and Technology, Karolinska Institutet, Stockholm, Sweden

2: Medical Department III - Endocrinology, Nephrology, Rheumatology, University of Leipzig Medical Center, Leipzig, Germany

3: Department of Diabetes and Endocrinology, Sheffield Teaching Hospitals, Sheffield, UK;

4: Department of Infection, Immunity and Cardiovascular Disease, The Medical School, The University of Sheffield, Sheffield, UK

5: Medical University of Innsbruck, Institute of Genetic Epidemiology, Innsbruck, Austria

6: Lausanne University Hospital, Centre de recherche clinique (CRC), Lausanne,

Switzerland

7: Charité-Universitätsmedizin Berlin, Department of Nephrology and Medical Intensive Care, Berlin, Germany

8: RWTH University of Aachen, Division of Nephrology and Clinical Immunology, Aachen, Germany

9: Department of Renal Medicine, University College London, London, UK

10: Sheffield Kidney Institute, Northern General Hospital, Herries Road, Sheffield, South

Yorkshire, UK

11: School of Health and Related Research, University of Sheffield, Sheffield, UK

Corresponding author:

Dr James Fotheringham

School of Health and Related Research

University of Sheffield

Regent Court, Regent Street, S1 4DA

Email: j.fotheringham@sheffield.ac.uk

*TE and NS contributed equally to this work.

Abstract

Rationale & Objective: Poor glycemic control may contribute to the high mortality rate in patients with type 2 diabetes receiving hemodialysis. Insulin type may influence glycemic control and its choice may be an opportunity to improve outcomes. This study assessed if treatment with analog insulin compared to human insulin is associated with different outcomes in people with type 2 diabetes and kidney failure receiving hemodialysis.

Study Design: Retrospective cohort study.

Setting & Participants: People in the Analyzing Data, Recognizing Excellence and Optimizing Outcomes (ARO) ii study with kidney failure commencing hemodialysis and type 2 diabetes being treated with insulin within 288 dialysis facilities between 2007 and 2009 across seven European countries. Study participants were followed for 3 years. People with Type 1 diabetes were excluded using an established administrative data algorithm.

Exposure: Treatment with an insulin analog or human insulin.

Outcomes: All-cause mortality, major adverse cardiovascular events, all-cause hospitalization, and confirmed hypoglycemia (blood glucose <3.0 mmol/l sampled during hemodialysis).

Analytical Approach: Inverse probability weighted Cox-proportional hazards models to estimate hazard ratios for analog insulin compared to human insulin.

Results: There were 713 insulin analog and 733 human insulin users. Significant variation in insulin type by country was observed. Comparing analog to human insulin at three years, the percentage of patients experiencing endpoints (and adjusted hazard ratios) were 22.0% vs 31.4% (0.808, 95%CI 0.66 to 0.99, p=0.04) for all-cause mortality, 26.8% vs 35.9% (0.817, 95% CI 0.68 to 0.98, p=0.03) for MACE, and 58.2% vs 75.0% (0.757, 95% CI 0.67 to 0.86, p<0.001) for hospitalization. Hypoglycemia was comparable between insulin types at 14.1% vs 15.0% (1.169,

95% CI 0.80 to 1.72, p=0.4). Consistent strength and direction of the associations were observed across sensitivity analyses.

Limitations: Residual confounding, lack of more detailed glycemia data.

Conclusions: In this large multinational cohort of people with type 2 diabetes and kidney failure receiving maintenance hemodialysis, compared to human insulin, treatment with analog insulins was associated with better clinical outcomes.

Index Words: Hemodialysis, mortality, major adverse cardiovascular events, Type 2 diabetes, insulin

Plain Language Summary

People with diabetes who are receiving dialysis for kidney failure are at high risk of cardiovascular disease and death. This study uses information from 1446 people from seven European countries who have kidney failure, are receiving dialysis, have type 2 diabetes, and are prescribed either insulin identical to that made in the body (human insulin) or insulins with engineered extra features (insulin analog). After three years, fewer participants receiving analog insulins had died, had been admitted to the hospital, or had a cardiovascular event (heart attack, stroke, heart failure or peripheral vascular disease). These findings suggest that analog insulins should be further explored as treatment leading to better outcomes for people with diabetes on dialysis.

Introduction

Diabetes is a major risk factor for developing chronic kidney disease (CKD) and the leading cause of kidney failure (KF) in the Western world.¹ According to the International Diabetes Federation (IDF) atlas,² the number of patients with diabetes will continue to rise from currently about 537 million to 643 million in 2030. Importantly, non-diabetic patients with KF have a very high mortality³ due to different pathomechanisms.^{4,5} Diabetes further increases the risk for all-cause and cardiovascular mortality,⁶ and patients with type 2 diabetes⁷ on hemodialysis (HD) have a very poor prognosis.

Importantly, patients with diabetes and KF show much higher rates of hypo- and hyperglycemic crises compared to non-KF high-risk populations with diabetes. Poor glycemic control in KF patients on HD⁸ or prior to dialysis initiation⁹ has been associated with higher mortality risk. As CKD progresses insulin clearance decreases¹⁰ leading to increased glycemic variability in patients with KF treated with exogenous insulin¹¹, driving insulin dosage adjustments. Exogenous human insulin (i.e. rapid- and long-acting) has several clinical shortcomings, including postprandial hyperglycemia followed by hypoglycemia and weight gain.¹² Consequently, rapid- and long-acting insulin analogues have been designed to better imitate the endogenous insulin response in healthy individuals.¹² Novel insulin analogues translate into better glycemic control¹³ and lower glycemic variability¹⁴ in the general population with diabetes. However, in clinical trials in these patients, no differences in mortality and major adverse cardiac events (MACE) have been found when comparing analogue and human insulins.¹⁵⁻¹⁷

Given that HD significantly contributes to an increased glycemic variability,¹⁸ the vulnerable nature of patients with type 2 diabetes on HD treatment,¹⁹ and the fact that CKD/KF represents a

major global disease burden in patients with diabetes,²⁰ there is a need for therapeutic improvements in the management of patients with type 2 diabetes receiving HD. To determine if analogue insulin compared to human insulin is associated with different outcomes in this group, we performed analyses of a large multicenter cohort of incident HD patient²¹ with type 2 diabetes from >250 Fresenius Medical Care (FMC) dialysis centers out of seven participating countries using causal inference techniques.

Methods

Cohort & Data

The Analyzing Data, Recognizing Excellence and Optimizing Outcomes (ARO) ii cohort was a prospective observational cohort study of incident HD patients enrolled at one of the 312 Fresenius Medical Care (FMC) facilities across 15 European countries between 2007 and 2009. Although the original cohort study had follow-up until 2014, patients under follow-up at this timepoint were small. For the current analysis follow-up ends in 2012. ARO used electronic medical records to capture anonymized longitudinal individual-level data.²² All local ethical and regulatory obligations concerning patient data for each of the 15 participating countries were met at the time of data collection, and the institutional review board from the Medical University of Innsbruck [EK-Nr. 1339/2020] has approved the current analysis. Informed consent was obtained from all patients by FMC Europe. For purposes of this study, analysis was limited to countries providing at least 10 HD patients with type 2 diabetes on insulin treatment.

Data on demography, comorbidity, laboratory, hospitalization, mortality, medications and individual HD sessions were available. The presence of six comorbid conditions were recorded using International Classification of Diseases (ICD) 10 codes from administrative data using existing schema (ischemic heart disease, congestive heart failure, cerebrovascular accident,

peripheral artery disease, chronic obstructive pulmonary disease, dysrhythmia).

Eligibility, Insulin Exposure and Follow-up

Participants in AROii were assessed for eligibility throughout their follow-up. They were considered eligible if they were identified as having type 2 diabetes, receiving either a human or analogue insulin, with analysis starting at the date of first prescription of these insulins while receiving dialysis, with the date of first dialysis being the start of follow-up for patients who were on insulin before starting HD. If a patient developed diabetes having already begun HD, follow-up would begin when insulin was first prescribed. Patients who were switched between analogue and human insulins during the follow up were excluded. Because ICD10 coding defines individuals with diabetes as "Insulin Dependent" or "Non-Insulin Dependent" rather than type 1 and type 2, an existing validated administrative data algorithm was applied to HD patients with the codes of E10 or E11 to differentiate between type 1 and type 2 diabetes using the combination of age at onset of diabetes, current age, previous diabetic ketoacidosis and insulin pump use as described recently.²³ Analogue or human insulin therapy, as well as concomitant oral antidiabetic medication (Table S1), were identified using the Anatomical Therapeutic Chemical (ATC) Classification System medication codes with review of the free text supplied with the medications to confirm correct assignment to the respective ATC groups and associated human or analogue insulin type (Table S2). Follow-up was three years, censored for transfer out of an FMC facility, transplantation, recovery of kidney function or change in dialysis modality. **Endpoints**

All-cause mortality was defined as death while receiving HD for kidney failure. Hospitalization, MACE and hypoglycemia were analyzed as time from first insulin prescription while receiving HD to first corresponding event. These endpoints have the competing events of

death while receiving HD, or the censoring events above. A hospitalization event was defined as an admission to hospital lasting at least one day. A MACE event was defined as hospitalization with the primary reason for admission corresponding to the ICD10 codes of coronary, cerebral, or peripheral arterial events, heart failure or cardiac arrest (Table S3). According to the joint position statement of the American Diabetes Association and the European Association for the Study of Diabetes,²⁴ a hypoglycemia event was defined as a laboratory glucose of <3.0 mmol/l (i.e. level 2 hypoglycemia). Glucose measurements were commonly performed with bloods performed to monitor HD and KF as part of routine clinical care at the beginning of HD, although glucose measurements performed at other times were available. We report the frequency of glucose monitoring using this method for each arm and by country (Table S4). The study did not have access to capillary glucose monitoring routinely used by patients to monitor glycemic control and did not capture if the participant had symptoms associated with any glucose measurement.

Statistical Methods

Numbers of missing data for each variable are shown in Table S5. To visualize the event rates stratified by insulin type, accounting for the competing risk of death we report cumulative incidence function graphs.²⁵ Proportional hazards models are reported using inverse probability weighting (IPW), and multiple imputation in accordance with best practices was performed.²⁶ First, multiple imputation of the variables included in the weighting and endpoint models, predicted using these variables and the endpoints, was undertaken using predictive mean matching-generated twenty datasets. This imputation method is more robust to assumptions around linearity.²⁷ Second, for each imputed dataset logistic regression was used to obtain probabilities of analogue insulin prescription using baseline covariates (age (<=50,>50-60,>60-

70,>70-80,>80), sex, six comorbid conditions [Table S5], albumin (<=35,>35 g/L), phosphate (<=0.8, >0.8-1.5, >1.5 mmol/L, calcium (<=2.1, >2.1-2.6, >2.6 mmol/L), hemoglobin (<100,100-120,>120 g/L), glycated hemoglobin (HbA1c, <=6%, >6-7,>7-8,>8-9,>9), erythropoiesis stimulating agents (because of their impact on red-cell turnover and therefore HbA1c, <2000, 2000 to <6000, 6000-<12000, >=12000 units per week), time on dialysis (<1 or >=1 year), and body mass index (BMI, <21, 2 unit increases then >35 kg/m²)). Probabilities from these logistic regression models are converted into weights from the reciprocal of the probabilities estimated from the baseline covariates to generate a well-balanced pseudopopulation (table S6). Stabilization of weights was not required (mean unstabilized weight 1.001, SD 0.431 using all twenty datasets). Comparing analogue to human insulin, we report the rate in person-years and the absolute proportion of patients experiencing the endpoints of allcause mortality, MACE, all-cause hospitalization, and hypoglycemia (< 3mmol/l) at three years using the weighted dataset, and finally perform adjusted Cox-proportional regressions reporting pooled estimates across the twenty imputed datasets. The inclusion of imputed adjustment variables in our second stage ensures that uncertainty around the imputed adjustment variables is accommodated in endpoint estimates and has the capacity to further reduce bias. In Table S7, we present relevant data from the multiple imputation process according to the standardized reporting guidelines adapted from Sterne et al.²⁸

We also conducted a number of subgroup analyses. Hazard ratios for cardiovascular and noncardiovascular mortality analyses are reported separately. We further estimated endpoints stratified by age, sex, BMI (<30 vs. ≥30 kg/m²), geography (Central/Eastern Europe: Czech Republic, Hungary, Turkey; Western Europe: United Kingdom, Italy, Portugal, Spain), reestimating the weights for these sub-populations (re-estimated weights reported in Table S8). To

explore if glucose variability differed between insulin types, we estimated the coefficient of variation for glucose measurements for each individual patient and estimated a mean coefficient of variation by insulin type. This analysis excluded patients from the United Kingdom and Portugal who had lower sampling rates than other participating countries. To explore if differences in outcomes could be explained by glycemic control, we present lowess smoothing plots of HbA1c for the duration of the analysis, stratified by insulin type. All analyses were undertaken in R version 4.1, with the packages "mice" and "ipw" to perform multiple imputation and IPW, respectively.

Results

Cohort and demography

Among 10,637 patients commencing HD across 15 countries, 3,783 were identified as diabetic subjects based on ICD10 codes of whom 1,899 were prescribed insulin prior to or while receiving HD. Of these, 239 were identified as type 1 diabetes using the administrative algorithm (Figure S1), and were therefore excluded from the analysis. A further 52 subjects were removed as they were from countries providing fewer than 10 patients, along with 162 patients who received a mixture of human and analogue insulins, leaving 1,446 patients recruited from 288 facilities across seven countries for the present analysis, with a total of 2855 patient-years follow-up (1.97 years per patient). Insulin was commenced prior to or within 90 days of starting HD in 69.8% (1,009 patients). The study flow diagram is shown in Figure S2. There was only a small proportion of patients (N=75) receiving oral antidiabetic medication at 90 days from starting HD.

The demography, comorbidities, and laboratory variables stratified by insulin type are shown in Table 1 and demonstrate older patients with more baseline comorbidities in the human insulin

group, but worse glycemic control in the analogue group. Although use of the two insulins was approximately evenly split across the entire cohort, there was significant variation across the seven countries (Table S9): the proportion of patients in each insulin group by country is shown by increasing gross domestic product per capita in Figure 1, suggesting no clear relationship.

Association between analogue vs human insulin and endpoints

Within the 1,446 patients included in the adjusted analysis, there were 387 deaths (173 cardiovascular, 186 non-cardiovascular, and 28 unknown), 965 first hospitalizations, 454 first MACE events, and 138 first hypoglycemic events (< 3mmol/l sampled on HD) over a median follow-up of 27.7 months. Cumulative Incidence Function plots stratified by analogue and human insulin are presented in Figure 2 with proportions experiencing events in Table 2. Analogue insulin was associated with superior event-free survival for the primary endpoints of all-cause mortality, MACE, and hospitalization. Hypoglycemia, restricted to countries in which it was assessed regularly at dialysis (United Kingdom and Turkey removed, remaining n=958), was comparable with both insulins.

Multi-variable adjustment for demography, biochemical parameters, and comorbidities using doubly adjusted IPW estimated the hazard ratios [95% confidence interval (CI)] for all-cause mortality of 0.808 [0.659-0.991] (p=0.04); MACE 0.817 [0.680-0.983] (p=0.03); hospitalization 0.757 [0.665-0.861] (p<0.001) and hypoglycemia 1.169 [0.796 – 1.718] (p=0.4) (Table 2). Cardiovascular and non-cardiovascular specific-mortality hazard ratios for analogue versus human insulin were 0.734 [0.541-0.996] (p=0.05) and 0.834 [0.617-1.128] (p=0.2) in the adjusted IPW analysis (Table 2). When the components of the MACE endpoint were analyzed separately (Table 2), results remained similar in terms of effect strength and direction. The mean coefficient of variation of glucose levels measured on HD, in countries performing it regularly,

was 0.328 for the analogue insulin group and 0.326 for the human insulin group (t-test p=0.9), and the trajectory of HbA1c in the two groups were not visually different (Figure 3).

Subgroup Analyses

Subgroup analyses showed consistent direction of effect in favor of analogue insulin for different subgroups of age, BMI, sex, and geographical region (Figure 4), with significant interaction between insulin type and geographical region for MACE and hospitalization (Table S8). Ethnicity was not captured by this study.

Discussion

Using a large, pan-European, multicenter cohort of incident HD patients with type 2 diabetes, we show that analogue insulin therapy is associated with lower all-cause mortality, MACE, and hospitalization compared to human insulin therapy, while level 2 hypoglycemia (<3.0 mmol/l) on HD and glycemic variability remained similar compared to human insulin.

Patients with KF treated by HD have a shortened lifespan with a difference of >25 years when compared to the general population.¹ In individuals with type 2 diabetes, KF is associated with an excess mortality rate compared to non-KF individuals with type 2 diabetes.²⁹ Importantly, patients with type 2 diabetes on HD are excluded from most of the cardio-renalprotective glucose-lowering medications, including sodium–glucose cotransporter 2 (SGLT2) inhibitors,³⁰ and insulin therapy is the cornerstone of antihyperglycemic treatment in KF.

In contrast to our findings in people with type 2 diabetes and KF on hemodialysis, in the general diabetic population, analogue compared to human insulin treatment is not associated with differences in mortality and MACE outcomes.¹⁵⁻¹⁷ Neugebauer et al.¹⁵ demonstrated similar rates of all-cause and cardiovascular mortality, as well as MACE, using a retrospective dataset from four US health care delivery systems comprising 127,600 adults with type 2 diabetes,

however the adjustment for time-varying HbA1c means that any advantage mediated through improved glycemic control would be neutralized. Fullerton et al.¹⁶ summarized data from randomized controlled trials on short-acting insulins and found no clear difference between analogue and human insulin, similar to a recent Cochrane review.¹⁶ There is no trial which has been designed to investigate long term effects, i.e. on mortality and MACE, in participants with diabetes-related microvascular complications, such as CKD. We believe that in general type 2 diabetes cohorts (not including people with CKD/KF) other factors and co-variates may counterbalance any potential differences between analogue vs. human insulin, for instance concomitant use of other (cardio-renal) protective glucose-lowering and/or lipid-lowering treatment. Notably, as statins do not decrease mortality and MACE in type 2 diabetes patients receiving HD,⁸ lipid-lowering treatment is unlikely to contribute to the observed effects.

Pathophysiologically, different mechanisms for both long- and short-acting analogues could potentially explain the improved outcome of patients on analogue insulins in our cohort. A recent meta-analysis in individuals with type 1 diabetes showed that analogue insulin therapy was associated with lower total, nocturnal, and severe hypoglycemia risk, as well as reduced postprandial glucose and HbA1c.³¹ In type 2 diabetes long-acting analogues do not result in insulin peaks compared to long-acting human insulins, i.e. neutral protamine Hagedorn insulin, thereby reducing hypoglycemia risk.³² In contrast, short-acting analogues induce a faster and higher peak plasma insulin concentration in the first hour after injection compared to human short-acting insulin, thereby reducing the adverse postprandial glucose peak.³³ Both, long- and short-acting, analogues, therefore, could significantly reduce glycemic variability, which has been linked to mortality in people on HD,³⁴ without necessarily modifying HbA1c. Two recent retrospective population-based cohort studies from UK and Taiwan have also reported beneficial

effects of (long-acting) analogue insulin with respect to cardiovascular outcomes, as well as hypoglycemic risk, supporting our findings in this more vulnerable cohort.^{35,36}

Cumulative incidence curves for mortality, MACE, and hospitalization outcome separated as early as the first months of follow-up, i.e. around initiation of HD. This is in accordance with observational data from the same cohort,³⁷ demonstrating an increased risk of hospitalization, MACE, and mortality in patients soon after the start of HD. Hypothetically, after patients have experienced an early outcome and dropped out, longitudinal trends stabilize over the study period and run in parallel (Figure 2) in our cohort. In addition, some patients already started analogue insulins before being included in our study and potential beneficial treatment effects of analogue insulins could have accumulated prior to starting HD.

Some limitations of the present study need to be highlighted. Although we have performed multivariable analyses adjusting for many clinically relevant covariates, residual confounding cannot be excluded. In particular we had restricted access to information on socioeconomic status of the participants, as increased costs of analogue insulin in some geographies may introduce bias. Even though the number of missing data for some variables was low, it required the use of multiple imputation. We did not have access to the more sensitive self-monitored blood glucose measurements to thoroughly assess glycemic variability. Because of the time-varying nature of insulin prescriptions it was not feasible to investigate different insulin regimens (e.g. basal insulin-only vs. basal-bolus insulin regimen) or different types of analogues (e.g. fast-vs. long-acting) separately. We acknowledge that excluding the small number of patients who switched between insulin types could introduce bias and does not utilize all available data. The study's strengths include the potentially high relevance for daily practice as there is an unmet need to improve the prognosis of individuals with type 2 diabetes and KF. Furthermore, we have

applied sophisticated statistical analyses with robust adjustment for multiple variables using a

validated clinical database, and analyzed many patients using well-defined outcomes.

Prospective, adequately designed and powered clinical trials should investigate whether

switching from human insulin to analogue insulin improves patient-relevant outcomes in people

with KF and the cost-effectiveness of this intervention.

In conclusion, in a large, multinational cohort of HD patients with type 2 diabetes, analogue

compared to human insulins were associated with better clinical outcomes, although

hypoglycemia rates were increased. Analogue insulins may represent a superior therapeutic

option for this group of patients with high unmet needs.

Supplementary Material

Supplementary File (PDF)

Figure S1: Decision tree for the identification of patients with type 1 diabetes (T1D) or type 2 diabetes (T2D) with patient numbers

Figure S2: Study flow diagram

Table S1: Overview on the Anatomical Therapeutic Chemical (ATC) classification system

 medication codes used to identify non-insulin glucose-lowering therapies

Table S2: Overview on the Anatomical Therapeutic Chemical (ATC) classification system medication codes used to identify analogue or human insulins, as well as pre-mixed insulins **Table S3:** Overview on the International Classification of Diseases (ICD) 10 codes used to identify comorbidities and complications

Table S4: Frequency of glucose monitoring according to country of the dialysis facility and insulin type, as well as insulin type stratified by each country

Table S5: Number of missing data for each variable stratified by the two study groups (i.e., human insulin vs. analogue insulin treatment)

Table S6: Baseline characteristics stratified by the two study groups (i.e. human insulin vs. analogue insulin treatment) in the observed population (presented in Table 1), as well as the inverse probability weighted pseudo-population

 Table S7: Multiple Imputation Reporting

Table S8: Summary statistics for inverse probability weights for subgroup analyses and interaction term p-values

Table S9: Demographics of analogue and human insulin users by country

Article Information

Additional Information: Dr Fotherinham's ORCiD is 0000-0002-8980-2223.

Authors' Contributions: Work conceptualization: TE, JaF; statistical analysis: TE, JaF, MG, NS; study design and data acquisition: CL, MF, KUE, JuF, FK, PS and DCW. Each author contributed important intellectual content during manuscript drafting or revision and agrees to be personally accountable for the individual's own contributions and to ensure that questions pertaining to the accuracy or integrity of any portion of the work, even one in which the author was not directly involved, are appropriately investigated and resolved, including with documentation in the literature if appropriate.

Support: Dr Stenvinkel's research benefited by support from the Strategic Research Program in Diabetes at Karolinska Institutet (Swedish Research Council grant No 2009-1068), Heart and Lung Foundation (20160384), Njurfonden and Westmans Foundation. Baxter Novum is a result of a grant from Baxter Healthcare to Karolinska Institutet. Dr Ebert was supported by a Novo Nordisk postdoctoral fellowship run in partnership with Karolinska Institutet, Stockholm, Sweden, a Karolinska Institutet Research Foundation grant, the Stiftelsen Stig och Gunborg Westman, the Deutsche Diabetes Gesellschaft (DDG), as well as by the Swedish Kidney Foundation (Njurfonden). Dr Ebert was further funded through the EFSD Mentorship Programme supported by AstraZeneca. Dr Floege received support from the German Research Foundation (SFB TRR 219, projects C1 and M1). Dr Fotheringham is supported by a National Institute for Health Research (UK) Clinician Scientist Award. The funders of this study had no role in study design, data collection and analysis, data interpretation, writing of the report or the decision to submit the paper for publication.

Financial Disclosure: The ARO CKD Research Initiative is a joint observational research commitment from Amgen and FMC (Europe), fully funded by Amgen (Europe), Rotkreuz, Switzerland. Dr Wheeler has received consultancy fees from Amgen, Astellas, AstraZeneca, Bayer, Boehringer Ingelheim, GlaxoSmithKline, Gilead, Janssen, Mundipharma, Napp Tricida, Vifor and Zydus. KE has received grants from Amgen, Astra Zeneca, Bayer, Evotec and Vifor, and honoraria from Akebia, Astra Zeneca, Bayer, Otsuka and Retrophin. Dr Floege has received honoraria from Amgen, AstraZeneca, Bayer, Boehringer, Fresenius, Vifor and serves on a data safety monitoring board in studies sponsored by Novo Nordisk. Dr Stenvinkel has served on scientific advisory boards of Baxter, Astra Zeneca, Baxter, Novo Nordisk, Fresenius and Pfizer/BMS. Dr Ebert has received honoraria from Sanofi, the CME-Verlag, and Santis. Dr Fotheringham has received honoraria from Fresenius, Novartis and Vifor Pharma and conducts research funded by Baxter and Boehringer Ingelheim. The remaining authors declare that they have no relevant financial interests.

Peer Review: Received November 7, 2022. Evaluated by 2 external peer reviewers, with direct editorial input from a Statistics/Methods Editor, an Associate Editor, and the Editor-in-Chief. Accepted in revised form May 23, 2023.

References

1. Johansen KL, Chertow GM, Gilbertson DT, et al. US Renal Data System 2021 Annual Data Report: epidemiology of kidney disease in the United States. *Am J Kidney Dis*. 2022;79(4)(suppl 1):A8-A12. doi: 10.1053/j.ajkd.2022.02.001

2. International Diabetes Federation. *IDF Diabetes Atlas: 10th Edition.* 2021. https://www.diabetesatlas.org

3. Jager dDJ, Grootendorst DC, Jager KJ, et al. Cardiovascular and Noncardiovascular Mortality Among Patients Starting Dialysis. *JAMA*. 2009/10/28 2009;302(16):1782-1789.

doi:10.1001/jama.2009.1488

4. Ebert T, Neytchev O, Witasp A, Kublickiene K, Stenvinkel P, Shiels PG. Inflammation and Oxidative Stress in Chronic Kidney Disease and Dialysis Patients. *Antioxidants & Redox Signaling*. 2021/12/10 2021;35(17):1426-1448. doi:10.1089/ars.2020.8184

5. Kalantar-Zadeh K, Jafar TH, Nitsch D, Neuen BL, Perkovic V. Chronic kidney disease. *The Lancet*. 2021/08/28 2021;398(10302):786-802. doi:10.1016/S0140-6736(21)00519-5

6. Fox CS, Matsushita K, Woodward M, et al. Associations of kidney disease measures with mortality and end-stage renal disease in individuals with and without diabetes: a meta-analysis. *The Lancet*. 2012/11/10 2012;380(9854):1662-1673. doi:10.1016/S0140-6736(12)61350-6

7. Wanner C, Krane V, März W, et al. Atorvastatin in Patients with Type 2 Diabetes Mellitus Undergoing Hemodialysis. *New England Journal of Medicine*. 2005/07/21 2005;353(3):238-248. doi:10.1056/NEJMoa043545

8. Hayashino Y, Fukuhara S, Akiba T, et al. Diabetes, glycaemic control and mortality risk in patients on haemodialysis: the Japan Dialysis Outcomes and Practice Pattern Study. *Diabetologia*. 2007/06/01 2007;50(6):1170-1177. doi:10.1007/s00125-007-0650-z

9. Rhee CM, Kovesdy CP, Ravel VA, et al. Association of Glycemic Status During Progression of Chronic Kidney Disease With Early Dialysis Mortality in Patients With Diabetes. *Diabetes Care*. 2017/08 2017;40(8):1050-1057. doi:10.2337/dc17-0110

10. Duckworth WC, Kitabchi AE. Insulin Metabolism and Degradation. *Endocrine Reviews*. 1981/04 1981;2(2):210-233. doi:10.1210/edrv-2-2-210

11. Iglesias P, Díez JJ. Insulin therapy in renal disease. *Diabetes, Obesity and Metabolism*. 2008/10 2008;10(10):811-823. doi:10.1111/j.1463-1326.2007.00802.x

12. Mathieu C, Gillard P, Benhalima K. Insulin analogues in type 1 diabetes mellitus: getting better all the time. *Nature Reviews Endocrinology*. 2017/04/21 2017;13(7):385-399. doi:10.1038/nrendo.2017.39

13. Mannucci E, Caiulo C, Naletto L, Madama G, Monami M. Efficacy and safety of different basal and prandial insulin analogues for the treatment of type 2 diabetes: a network meta-analysis of randomized controlled trials. *Endocrine*. 2021/12/01 2021;74(3):508-517. doi:10.1007/s12020-021-02889-6

14. Pérez-Maraver M, Caballero-Corchuelo J, Boltana A, Insa R, Soler J, Montanya E. Comparison of human insulin and insulin analogues on hypoglycaemia and metabolic variability in type 1 diabetes using standardized measurements (HYPO score and Lability Index). *Acta Diabetol.* 2013/08/01 2013;50(4):529-535. doi:10.1007/s00592-011-0320-y

15. Neugebauer R, Schroeder EB, Reynolds K, et al. Comparison of Mortality and Major Cardiovascular Events Among Adults With Type 2 Diabetes Using Human vs Analogue Insulins. *JAMA Netw Open*. 2020/01/03 2020;3(1):e1918554-e1918554. doi:10.1001/jamanetworkopen.2019.18554

16. Fullerton B, Siebenhofer A, Jeitler K, et al. Short-acting insulin analogues versus regular human insulin for adult, non-pregnant persons with type 2 diabetes mellitus. *Cochrane Database of Systematic Reviews*. 2018;(12):CD013228. doi:10.1002/14651858.CD013228

17. Price HI, Agnew MD, Gamble J-M. Comparative cardiovascular morbidity and mortality in patients taking different insulin regimens for type 2 diabetes: a systematic review. *BMJ Open*. 2015/03/01 2015;5(3):e006341. doi:10.1136/bmjopen-2014-006341

18. Abe M, Kalantar-Zadeh K. Haemodialysis-induced hypoglycaemia and glycaemic disarrays. *Nature Reviews Nephrology*. 2015/04/07 2015;11(5):302-313. doi:10.1038/nrneph.2015.38

19. Galindo RJ, Ali MK, Funni SA, et al. Hypoglycemic and Hyperglycemic Crises Among U.S. Adults With Diabetes and End-stage Kidney Disease: Population-Based Study, 2013–2017. *Diabetes Care*. 2021/11/05 2021;45(1):100-107. doi:10.2337/dc21-1579

20. Bikbov B, Purcell CA, Levey AS, et al. Global, regional, and national burden of chronic kidney disease, 1990–2017: a systematic analysis for the Global Burden of Disease Study 2017. *The Lancet*. 2020/02/29 2020;395(10225):709-733. doi:10.1016/S0140-6736(20)30045-3

21. Ebert T, Qureshi AR, Lamina C, et al. Time-dependent lipid profile inversely associates with mortality in hemodialysis patients – independent of inflammation/malnutrition. *Journal of Internal Medicine*. 2021 2021;290(4):910-921. doi:10.1111/joim.13291

22. Floege J, Gillespie IA, Kronenberg F, et al. Development and validation of a predictive mortality risk score from a European hemodialysis cohort. Article. *Kidney International*. May 2015;87(5):996-1008. doi:10.1038/ki.2014.419

23. Weisman A, Tu K, Young J, et al. Validation of a type 1 diabetes algorithm using electronic medical records and administrative healthcare data to study the population incidence and prevalence of type 1 diabetes in Ontario, Canada. *BMJ Open Diabetes Research & Care*. 2020;8(1):e001224. doi:10.1136/bmjdrc-2020-001224

24. The International Hypoglycaemia Study Group. Glucose concentrations of less than 3.0 mmol/l (54 mg/dl) should be reported in clinical trials: a joint position statement of the American Diabetes Association and the Europian Association for the Study of Diabetes. *Diabetologia*. 2017/01/01 2017;60(1):3-6. doi:10.1007/s00125-016-4146-6

25. Austin PC, Lee DS, Fine JP. Introduction to the Analysis of Survival Data in the Presence of Competing Risks. *Circulation*. 2016;133(6):601-609.

doi:doi:10.1161/CIRCULATIONAHA.115.017719

26. Leyrat C, Seaman SR, White IR, et al. Propensity score analysis with partially observed covariates: How should multiple imputation be used? *Statistical Methods in Medical Research*. 2019;28(1):3-19. doi:10.1177/0962280217713032

27. Schenker N, Taylor JMG. Partially parametric techniques for multiple imputation. *Computational Statistics & Data Analysis*. 1996;22(4):425-446.

28. Sterne JAC, White IR, Carlin JB, et al. Multiple imputation for missing data in epidemiological and clinical research: potential and pitfalls. *BMJ*. 2009/06/29 2009;338:b2393. doi:10.1136/bmj.b2393

29. Morton JI, Sacre JW, McDonald SP, Magliano DJ, Shaw JE. Excess all-cause and causespecific mortality for people with diabetes and end-stage kidney disease. *Diabetic Medicine*.e14775. doi:10.1111/dme.14775

30. American Diabetes AssociationProfessional Practice Committee. 11. Chronic Kidney Disease and Risk Management: Standards of Medical Care in Diabetes—2022. *Diabetes Care*. 2021;45(Supplement_1):S175-S184. doi:10.2337/dc22-S011

31. Melo KFS, Bahia LR, Pasinato B, et al. Short-acting insulin analogues versus regular human insulin on postprandial glucose and hypoglycemia in type 1 diabetes mellitus: a systematic review and meta-analysis. *Diabetology & Metabolic Syndrome*. 2019/01/03 2019;11(1):2. doi:10.1186/s13098-018-0397-3

32. Riddle MC, Rosenstock J, Gerich J. The treat-to-target trial: randomized addition of glargine or human NPH insulin to oral therapy of type 2 diabetic patients. *Diabetes Care*. Nov 2003;26(11):3080-6. doi:10.2337/diacare.26.11.3080

33. Mathieu C, Martens P-J, Vangoitsenhoven R. One hundred years of insulin therapy. *Nature Reviews Endocrinology*. 2021/12/01 2021;17(12):715-725. doi:10.1038/s41574-021-

00542-w

34. Shi C, Liu S, Yu HF, Han B. Glycemic variability and all-cause mortality in patients with diabetes receiving hemodialysis: A prospective cohort study. *J Diabetes Complications*. Apr 2020;34(4):107549. doi:10.1016/j.jdiacomp.2020.107549

35. Brunetti VC, Yu OHY, Platt RW, Filion KB. The association of long-acting insulin analogue use versus neutral protamine Hagedorn insulin use and the risk of major adverse cardiovascular events among individuals with type 2 diabetes: A population-based cohort study. *Diabetes, obesity & metabolism.* Nov 2022;24(11):2169-2181. doi:10.1111/dom.14802

36. Yang C-T, Li K-Y, Yang C-Y, Ou H-T, Kuo S. A nationwide cohort study for comparative vascular safety of long-acting insulin analogue versus intermediate-acting human insulin in type 2 diabetes. *Scientific Reports*. 2021/02/18 2021;11(1):4152. doi:10.1038/s41598-021-83253-6

37. Eckardt K-U, Gillespie IA, Kronenberg F, et al. High cardiovascular event rates occur within the first weeks of starting hemodialysis. *Kidney International*. 2015/11/01 2015;88(5):1117-1125. doi:10.1038/ki.2015.117

Journal Prerk

	Human insulin	Analog insulin
N	733	713
Age at baseline (years)	68.6 (10.2)	64.7 (11.2)
Male sex (%)	439 (59.9)	425 (59.6)
Vintage (days)	164 (357)	187 (367)
BMI (kg/m²)	27.4 (5.1)	28.1 (5.6)
Albumin (g/l)	37.0 (5.3)	37.0 (4.9)
Calcium (mmol/l)	2.2 (0.2)	2.2 (0.2)
Hemoglobin (g/l)	108 (16)	109 (16)
Phosphate (mmol/l)	1.4 (0.5)	1.5 (0.4)
HbA1c (%) / (mmol/mol)	7.1 (1.5) / 54.1 (1.7)	7.4 (1.8) / 57.4 (1.9)
ESA dosage (U/week)	4255 (7597)	4820 (6578)
Comorbidities		
Atherosclerotic heart disease (%)	232 (32.1)	152 (21.3)
Congestive heart failure (%)	184 (25.4)	98 (13.8)
Chronic obstructive pulmonary disease (%)	63 (8.7)	42 (5.9)
Stroke (%)	136 (18.8)	56 (7.9)
Dysrhythmias (%)	107 (14.8)	59 (8.3)
Peripheral artery disease (%)	168 (23.2)	98 (13.8)
Country		
United Kingdom (%)	23 (3.1)	81 (11.4)
Italy (%)	10 (1.4)	27 (3.8)
Spain (%)	165 (22.5)	230 (32.3)
Portugal (%)	349 (47.6)	35 (4.9)
Hungary (%)	49 (6.7)	16 (2.2)
Czech Republic (%)	116 (15.8)	15 (2.1)
Turkey (%)	21 (2.9)	309 (43.3)
Insulin after commencing HD (%)	208 (28.4)	229 (32.1)

Table 1. Baseline characteristics stratified by the two study groups (ie, human insulin vs. analog insulin treatment).

BMI, Body mass index; ESA, Erythropoiesis-stimulating agents; HbA1c, Glycated hemoglobin A1c; HD, Hemodialysis.

Data are presented as mean (standard deviation) for continuous measures, and N (percentage per insulin group) for categorical measures.

	Percent with event at 3 years (analog vs. human)	Event Rate per 100 patient years (analog vs. human)	Crude analysis	Adjusted IPW analysis
All-cause mortality	22.0 vs. 31.4	11.4 vs 15.6	0.73 (0.60 – 0.89)	0.81 (0.66 – 0.99)
MACE	26.8 vs. 35.9	16.1 vs 21.7	0.74 (0.62 – 0.89)	0.82 (0.68 – 0.98)
Coronary	9.0 vs 14.3	4.9 vs 7.9	0.62 (0.46 – 0.85)	0.74 (0.55 – 1.00)
Cerebral	7.2 vs 9.7	3.9 vs 5.0	0.77 (0.53 – 1.10)	0.90 (0.62 - 1.29)
Heart Failure/ Fluid Overload	6.0 vs 9.5	3.3 vs 5.0	0.65 (0.44 to 0.95)	0.81 (0.55 - 1.18)
Peripheral	5.9 vs 14.2	3.2 vs 7.8	0.41 (0.29 – 0.59)	0.45 (0.32 - 0.64)
Hospitalization	58.2 vs. 75.0	50.0 vs 73.2	0.70 (0.61 – 0.79)	0.76 (0.67 – 0.86)
Hypoglycemia	14.1 vs. 15.0	7.6 vs 6.5	1.18 (0.81 – 1.72)	1.169 (0.80 – 1.72)
CV mortality	9.6 vs. 14.8	4.9 vs 7.2	0.68 (0.50 – 0.92	0.73 (0.54 – 0.99)
Non-CV mortality	11.0 vs. 15.2	5.6 vs 7.4	0.76 (0.56 – 1.01)	0.83 (0.62 – 1.13)

Table 2: Absolute and relative risks associated with analog insulin compared to the human insulin group and endpoint.

CV, Cardiovascular; MACE, Major adverse cardiac events. All other abbreviations as indicated in Table 1.

Absolute risks, event rat, as well as weighted using inverse probability weighting (crude analysis) and multivariate (right column) hazard regression analysis adjusted for demography (BMI, albumin, phosphate, calcium, hemoglobin, HbA1c, ESA usage, time on dialysis) and comorbidities (ischemic heart disease, heart failure, chronic obstructive pulmonary disease, cerebrovascular disease, dysrhythmia, peripheral artery disease) using doubly adjusted inverse probability weighting (IPW). Hazard ratios and 95% confidence intervals are depicted for the analogue insulin group compared to the human insulin group (reference group).

Figure 1: Proportion of patients in each insulin group stratified by country of the dialysis facility. Human insulin is red, Analogue Insulin is teal. Countries are depicted by increasing gross domestic product per capita from left to right. Absolute numbers of patients included in the presented cohort is depicted on the y-axis.

Figure 2: Cumulative incidence function plots for the endpoints of all-cause mortality, cardiovascular (CV) mortality, non-CV mortality, MACE, all-cause hospitalization, and hypoglycemia (<3mmol/l, sampled on HD). Follow-up time is censored at 3 years by transplantation, loss to follow-up, transfer out of the dialysis facility, recovery of kidney function or change in dialysis modality. The competing event of mortality (non-CV mortality in the case of CV mortality) is displayed to the right. Human insulin is red, Analogue Insulin is teal.

Figure 3: Subgroup analyses for analogue insulin treatment compared to human insulin treatment (reference) according to age, body mass index (BMI), sex, and geographical region of the respective dialysis facility. Overall adjusted hazard ratios (circles) and 95% confidence intervals (lines) for analogue compared to human insulin are depicted in the bottom panels and are also shown in Table 2. In the four upper subgroup panels, black and grey circles/lines indicate the respective subgroups for the sensitivity analyses. All analyses were based on multivariable hazard regression analyses adjusted for demography, biochemical parameters (body mass index, albumin, phosphate, calcium, hemoglobin, glycated hemoglobin A1c, Erythropoiesis-stimulating agents usage, time on dialysis), and comorbidities (ischemic heart disease, heart failure, chronic obstructive pulmonary disease, cerebrovascular disease, dysrhythmia, peripheral artery disease) using doubly robust inverse probability weighting (IPW). MACE, Major adverse cardiac events.

Figure 4: Lowess smoothing plot of HbA1c values during the follow-up period stratified by use of Analogue and Human Insulin. Human insulin is red, Analogue Insulin is teal.







